CSCI 3194 (Professional, Ethics, and Design Seminar), Spring 2004

Final Group Evaluation

Name and Group Number:

Describe the work you did on this part of the project.

How well do you think your group did with this part of the project? What did you do well? What did you not do well? What if anything would you do differently if you were starting over?

Evaluate the contributions of the other members of your group. For each other member, did he or she attend group meetings and participate? contribute to the project? work well with other group members? Give each person a numerical rating (with 1 the best rating and 5 the worst) and comment if you like.

Name Rating Comments